Contents lists available at ScienceDirect



European Economic Review

journal homepage: www.elsevier.com/locate/eer



Long-lasting effects of indoctrination in school: Evidence from the People's Republic of Poland $\stackrel{\text{\tiny{$12}}}{\sim}$



Joan Costa-Font^{a,1}, Jorge García-Hombrados^{b,*}, Anna Nicińska^{c,2}

^a London School of Economics and Political Science (LSE), UK

^b Universidad Autónoma de Madrid, Spain

^c University of Warsaw, Poland

ARTICLE INFO

Dataset link: Source files (Original data)

JEL classification: 128

Keywords: Education systems Communist education Education reforms School curriculum Later life outcomes Human capital attainment Labour market participation

1. Motivation

ABSTRACT

This paper studies the effect of communist indoctrination in school on labour force participation and human capital investments. Specifically, we evaluate the impact of a reform in Poland that revoked political indoctrination in school in the mid-1950s, while leaving the rest of the curriculum unchanged. To overcome endogeneity concerns, we exploit cut-off birth dates for school enrolment that exhibit variation in the level of exposure to the reform. We find that a reduction in school indoctrination increased the probability of finishing secondary and tertiary education, and expanded labour force participation about 50 years down the line.

To date, more than 3 billion people worldwide live under the rule of autocrats.³ Autocratic regimes restrict the freedom of their compatriots and, to ensure regime survival, they often draw on repression and mass surveillance (Acemoglu and Robinson, 2005; Davenport, 2007; Arendt, 1951; Davenport and Armstrong, 2004), with lasting negative consequences in terms of development goals (Lichter et al., 2020; Acemoglu et al., 2020; Nikolova et al., 2019). A recurrent instrument to reduce political dissidence and nurture regime loyalty is the use of indoctrination in schools to instil obedience and consolidate a set of beliefs to be accepted uncritically (Anderson, 1994; Gerovitch, 2007). However, while school indoctrination can serve as a short-term tool to improve trust in government (Cantoni et al., 2017), the repression of individual sense of agency and critical thinking that encompasses educational systems in autocratic regimes may exert adverse long-lasting impacts on the long-term welfare of individuals (Gorodnichenko and Roland, 2011a,b).

https://doi.org/10.1016/j.euroecorev.2023.104641

Received 27 December 2022; Received in revised form 10 November 2023; Accepted 14 November 2023

Available online 18 November 2023

 $[\]stackrel{\circ}{\sim}$ The research was supported by the Polish National Science Centre from grant no. 2019/33/B/HS4/00387 and from the Polish National Agency for Academic Exchange Bekker programme grant no. PPN/BEK/2019/1/00039. We are grateful to Michał Brzeziński, Jose Montalban Castilla, Andrés Barrios, Paolo Masella, Leszek Morawski, Michał Myck, Paweł Strawiński, two anonymous referees and an associate editor for their valuable comments. All errors are our own and the usual disclaimer applies.

^{*} Correspondence to: Facultad de Ciencias Economicas, Universidad Autonoma de Madrid, Calle Francisco Tomas y Valiente n5, Cantoblanco, Madrid, Spain. *E-mail addresses:* j.costa-font@lse.ac.uk (J. Costa-Font), jorge.garciah@uam.es (J. García-Hombrados), anicinska@wne.uw.edu.pl (A. Nicińska).

¹ London School of Economics and Political Science (LSE), Houghton Street, WC2A 2AE, London, England, UK.

² Faculty of Economic Sciences & Digital Economy Lab University of Warsaw, Długa 44/50, 00-241, Warsaw, Poland.

³ Information gathered from Our World in Data database, publicly available in the following link: https://ourworldindata.org/democracy.

^{0014-2921/© 2023} The Authors. Published by Elsevier B.V. This is an open access article under the CC BY-NC license (http://creativecommons.org/licenses/by-nc/4.0/).

J. Costa-Font et al.

This paper investigates the long-term effects of school indoctrination on human capital accumulation and labour force participation. To this end, we assess the impact of a reform that removed Marxism–Leninism indoctrination from the school system of the former People's Republic of Poland in the 1954/55 school year. The remainder of the curriculum and system of education remained unaltered by the reform.

To estimate the causal effects of the education reform, we use census data and exploit the variation in exposure to the reform induced by individuals' dates of birth. More specifically, we compare those who were born a few weeks ahead of or after the annual cutoff date for starting school every year. In communist Poland, children turning 7 before December 31st of a given calendar year were enrolled in school starting from September of the same year. Thus, among individuals who were already in school when the reform was implemented, those born just after the cut-off date to initiate school in a given year (January to March), were exposed to one additional year of post-reform education relative to individuals who were born in the last weeks of the previous calendar year (October to December), despite being only a few weeks younger.

To net out the effect of exposure to one additional year of post-reform education from the effect of starting school at an older age, or from seasonality effects in the month of birth, we use a difference-in-differences strategy comparing individuals born just before and after the cut-off date among two groups of people. The first group includes individuals who were already enrolled in compulsory education at the time of the reform. For these individuals, being born in the first quarter of the year led to one additional year of post-reform education during compulsory education relative to those individuals born in the last quarter of the previous calendar year. The second group includes individuals who initiated school after the introduction of the reform. These group of individuals, were born in the first quarter of the year or in the last quarter of the previous calendar year, and pursued all their education in the post-reform educational system.

The results show that revoking political indoctrination in school exerted a positive effect on human capital attainment and labour force participation nearly 50 years later. We find that, relative to a year of education in the pre-reform (indoctrinating) education system, exposure to a year of post-reform education (that had abolished political indoctrination) increased the probability of completing secondary education, tertiary education, and labour force participation by 1.69, 1.15, and 1.42 percentage points, respectively. We interpret these estimates as strong evidence for the negative effects of Marxism–Leninism indoctrination on human capital investments and labour force participation. The main conclusions of the study are robust to the use of an alternative dataset, and a battery of robustness and placebo tests.

We then explore what potential mechanisms are driving the latter results. While we cannot perfectly disentangle which mechanisms channel the effect, our results suggest that exposure to the school reform in Poland led to a change in individuals' sense of agency, which has been documented to improve economic outcomes (Gorodnichenko and Roland, 2012, 2011b; Viinikainen et al., 2010).

The reform was implemented in September 1954 by a communist regime that survived until the late 1980s, and changed the values that had guided the educational system during the Stalinist period. The reform removed the contents dedicated to political indoctrination, which had consequences for the grading of pupils' school performance. Although the grading rules remained the same in the new school system, political activities such as Stalin-themed recitation competitions or contents explicitly praising the importance of obedience to the Soviet regime and adherence to Marxist–Leninist values were removed from schools. The new curriculum was also less partisan regarding individual agency and Western cultural contents, and children becoming members of communist youth organisations were no longer rewarded with better grades. More importantly, the assignments and tasks on which students were evaluated increased the importance of scholarly performance as opposed to demonstrations of the regime's support. By removing political indoctrination from both school curricula and grading, the reform encouraged students to shift their efforts into scholarly activities. All things considered, the reform replaced a curriculum that encouraged and reinforced passivity, fear, collectivism, obedience, and subordination with one that encouraged ambition, curiosity, critical thinking, and self-direction.

A crucial feature of our natural experiment is that the change in the educational system introduced by the reform was limited to the indoctrination contents in the curriculum and, as a result, also in the grading system. In other words, not only ideological contents and activities were removed from the school curriculum, but also any participation in these activities, alongside uncritical demonstrations of dedication to Marxism–Leninism, which were no longer academically rewarded. The rest of the school curriculum, evaluation system, and number of compulsory years of school remained the same. Nonetheless, abolishing political indoctrination in schools resulted however in three hours less of schooling per week and the time liberated was not substituted with instruction of other subjects. Hence, we examine whether the reduction in the instruction time could be driving the effects of the reform, we document no evidence supporting this hypothesis.

The closest paper to ours is Fuchs-Schündeln and Masella (2016). Using an identification strategy based on school entry cutoff dates in East Germany, Fuchs-Schündeln and Masella (2016) show that exposure to an additional year of education in the post-communist education system exerted a positive effect on both human capital accumulation and income relative to a year of education in the school system of the communist Democratic Republic of Germany. To the best of our knowledge, this is the only paper examining the causal effects of communist education. However, we believe our study makes two contributions relative to Fuchs-Schündeln and Masella (2016).

Firstly, we take into consideration that communist and non-communist education systems are different above and beyond their ideological focus (e.g. focus on technical and physical education, homework load, or approach to religion, among others). The paper by Fuchs-Schündeln and Masella (2016) does not disentangle the effect of indoctrination from other features of communist education, or other changes that may arise with broader school reforms in the context of a newly incepted country after the II World War, which was the case of the post-communist education reform in East Germany. In contrast, the Polish natural experiment that we use in this study allows examining the effect of a reform that abolished political indoctrination in schools in the 1950s. This is a

unique setting to identify the effects of Marxist-Leninist indoctrination and, especially, to disentangle its effect from other aspects of communist education. While the Polish reform also took place in a time of profound changes, the empirical strategy used and the reform examined in our paper exhibit some key advantages relative to other settings. Namely, they allow to isolate the effect of indoctrination from other features of communist education, and even other changes that may arise with broader school reforms.

Secondly, most studies assessing the effect of communist institutions focus on East Germany (see for example Alesina and Fuchs-Schündeln, 2007; Lichter et al., 2020; Fuchs-Schündeln and Masella, 2016). Our study focuses on Poland instead, a country closer to the Union of Soviet Socialist Republics (USSR) and, at the time, with considerably less exposure to Western culture than East Germany.

We contribute to different strands of the literature. Firstly, existing literature has documented that exposure to communist regimes can exert detrimental welfare consequences. Previous studies have investigated the role of different instruments used by autocratic regimes, including mass surveillance or the education system (Abadie et al., 2015; Lichter et al., 2020; Fuchs-Schündeln and Masella, 2016). This paper adds to this literature by specifically isolating the role of political indoctrination in school, one of the potential mechanisms that could contribute to explaining the adverse effects of exposure to communism, and communist education, on welfare. Furthermore, our study expands the geographical focus by investigating a policy reform in Poland, which adds to the existing research on the effects of communism which so far refers primarily to East Germany.

Secondly, the article extends the growing literature that investigates several aspects of school curriculum that shape the long-term effects of education (Görlitz and Gravert, 2018; Rose and Betts, 2004; Bernheim et al., 2001; Benavot, 1992; Arold, 2022). We show that both the school curricula contents and grading exert long-lasting consequences on human capital accumulation and labour market outcomes.

Finally, this article adds to the growing body of evidence on the long-term consequences of values and attitudes (Gorodnichenko and Roland, 2011a,b, 2012; Hansen, 2013; Granato et al., 1996; Aspachs-Bracons et al., 2008). Previous evidence has shown that communist education systems (and, more generally, communist regimes) successfully instil certain attitudes and values (Cantoni et al., 2017; Alesina and Fuchs-Schündeln, 2007). While we cannot perfectly identify all the specific mechanisms driving the effect of the reform, our results suggest that the promotion of individual agency could explain its beneficial effects on crucial welfare determinants, such as labour force participation and human capital accumulation.

The main values promoted by Marxism–Leninism are collectivism and anti-capitalism. Whilst Marxism–Leninism does not demote education and labour force participation, it promotes obedience to the regime and the repression of individual agency, which may ultimately affect these outcomes (Gorodnichenko and Roland, 2011a,b). The promotion of obedience and demotion of individual sense of agency are standard features of school systems in autocratic regimes (Diwan and Vartanova, 2020). Thus, the suggestive evidence regarding the role of values associated with individual agency presented in this paper opens the debate as to whether political indoctrination in schools in other autocratic regimes could lead to similar detrimental effects.

The structure of the paper is the following. Section 2 discusses the institutional setting. Section 3 describes the data and Section 4 introduces the empirical strategy and discusses its validity. Section 5 presents the main results of the study and Section 6 examines what potential mechanisms could be driving the effects of the reform. Section 7 examines the robustness of the main results to different placebo and empirical checks. Section 8 concludes.

2. The education system in Poland

2.1. The education system in Poland before the 1954/55 reform

The Republic of Poland gained political independence in 1918 and efficiently introduced free compulsory education lasting seven years throughout the entire country by 1920 (Kurian, 1988). In September 1938, the school year that free compulsory education was incepted, already 91.2% of children aged 7 were enrolled in primary education and compliance with compulsory education was 90.6% (Mauersberg, 1996). After 1939, both the Nazis and Soviets occupying Polish lands introduced their educational settlements. However, during the post-war period, Poland, as a member of the Soviet bloc, adopted a communist education model based on free compulsory education in secular public schools tightly controlled by central level authorities.

Education played a crucial role in Soviet regimes by stimulating the transmission of socialist values, aiming at the construction of the "new Soviet man" (Anderson, 1994; Gerovitch, 2007; Hoisington et al., 2000). To achieve this goal, Marxist-Leninist ideological content was efficiently introduced into Polish schools all the way from the school year 1948/49 (Jankowiak, 2001) when 98.9% of children aged 7–10 complied with compulsory education laws (Mauersberg and Walczak, 2005).⁴ Schools served as means of indoctrination aimed at diminishing individual agency, instilling obedience and the values of subordination, atheism, anti-religion, and anti-capitalism (Jarosz, 1998). The school curriculum reflected these values by illustrating school knowledge with examples of communism superiority and capitalistic exploitation, as well as elements of 'historical materialism'.⁵ Moreover, the value of high culture (referred to as bourgeois) was diminished, Western pop culture was absent, and local folklore culture had a substantial presence in school curricula (e.g. music in art classes and dance in physical education). Obligatory reading lists (e.g. including books by Dickens, Twain, Sienkiewicz or Orzeszkowa, among others) comprised entirely of books that emphasised inequality,

⁴ The instability of education system in early-1940s and physical destruction of schools' staff and infrastructure in the mid-1940s hardly lends the opportunity to examine effects of the introduction of ideological indoctrination to Polish schools.

⁵ For example, in primary school, pupils heard about an American millionaire who bequeathed his wealth, accumulated through the exploitation of workers, to his dog, while many children in his neighbourhood lived in starkly depicted poverty (Wagner, 2018).

poverty, and immorality of capitalist relations rooted in competition rather than cooperation (Jedrych, 2014). All school subjects contained explicit praise of the USSR in general and of Stalin in particular. General school assemblies praising Marxism–Leninism were organised regularly during school hours, in which active or passive participation was compulsory for all students.

2.2. The 1954/55 school reform

The education reform examined in this paper drastically and almost instantaneously abolished political indoctrination in school.⁶ Starting from the 1954/55 school year, indoctrination was revoked from courses of Polish, Russian, history, constitution, mathematics, chemistry, biology, geography, astronomy, logic, and physical education in all schools in Poland (Dobosiewicz, 1971). Once education ceased to be mainly the way of "inculcating in the young generation a Marxist-Leninist world view", the grading of pupils performance was to a larger degree dictated by their knowledge and skills, whereas before, a declared system of beliefs was a substantial factor in the pupils' assessment (Cary, 1976).⁷ As a result of the changes introduced by the reform, the school environment altered from one repressing individual agency and rewarding obedience, collectivism, and subordination to the new one, where self-direction, critical thinking, and some individualism were allowed (Ignaczak et al., 2019).

Examples of changes in the new educational curriculum include, among others, the removal of ideological activities such as Stalin-themed recitation competitions, mathematical relations no longer had to be illustrated with the number of Soviet army soldiers, physical education classes ceased to be an explicit way of shaping students' bodies accordingly to ideological expectations of Soviet authorities, and less attention was paid to books representing Marxist-Leninist literature in class (Tomasik, 2016). An anecdote that illustrates some of these changes occurred when the Minister of Education addressed the pupils as individuals in direct words 'beloved children, dear youth'⁸ in his speech opening the 1954/55 school year, while year earlier he had treated them as the members of Soviet organisations and spoken in communist jargon⁹ (Wagner, 2018). With the end of Stalinism, the education system continued to criticise capitalist relations explicitly but was no longer subjected to direct Marxist-Leninist indoctrination. The reform made education less partisan regarding religion, individualism, and Western cultural contents (Dobosiewicz, 1971), and shifted focus towards knowledge transmission and skills formation.

School indoctrination and the formation of values is not only defined by values taught explicitly in textbooks, but also by the grading system in place. Children willing to become members of communist youth organisations were no longer rewarded with better grades in school. In the reformed system, children's performance was assessed based on academic outcomes to a substantially greater degree than before, when participation in ideological activities was part of the school curriculum. While efforts put into education paid off for students in the new educational system, the uncritical participation in ideological activities was rewarded most before the reform (Świda-Ziemba, 1998). On top of that, students were no longer punished for a lack of explicit enthusiasm towards communism, and assignments allowed for responses formulated with children's own words (instead of repeating formulas to be learned by heart), giving them slightly more freedom of expression (Krzywicki, 1955). After the reform, children were allowed to be unpolitical individual creatures (Stańczyk, 2019) rather than encouraged to join the communist collective youth (Zysiak, 2016). The classrooms had more space for individual development, curiosity, and ambition.

Finally, the reform reduced the overall number of school hours from 29 to 26 per week. Such teaching time reduction resulted entirely from the removal of indoctrination contents. No other changes were introduced in the school curricula, school year duration, or the number of compulsory years of education (Dobosiewicz, 1971). In Section 4 we discuss whether the main conclusions of the study could be driven by the reduction in the number of school hours and we find evidence suggestive that this is not the case.

3. Data

The main analysis is conducted using the 2002 Polish census database. The 2002 Polish census wave collected crucial demographic and socio-economic information for the first time since the collapse of the communist regime in 1989. The population data collection was implemented in less than three weeks during the spring of 2002 and was addressed to both temporary and permanent residents as well as those temporarily abroad. However, the census database available online and used in this study provides information on a random sample of 3,823,596 individuals living in Poland. Unfortunately, the Polish censuses conducted in 1978, 1988, and 2011 are not included in the analysis because they do not provide information on the month and year of birth.¹⁰ The main advantage of the 2002 census is its large sample size.However, the census lacks information on income and values, although it provides information on labour force participation and educational attainment.

⁶ A critique of the influence of Marxism–Leninism on the Polish education system was initiated in 1952. The Magazine *Nowa Kultura* published an article entitled 'Letter to Miecia' in the form of a fictional letter from a student, complaining about the history class to the Minister of Education. This article sparked a discussion in two other influential journals, which eventually led to the Ministry of Education announcing changes to the curriculum in May 1954, to be implemented starting from the 1954/55 school year.

⁷ Cary (1976) gives examples of assignments such as "Tell about the fight of communist parties against revisionism, dogmatism, and sectarianism" or "Why is our motherland called socialist?" A proper answer to an assignment was scripted by the state officials to be learned by heart and repeated uncritically by pupils.

⁸ Originally 'Kochane dzieci. Droga młodzieży'; authors' own translation from Polish to English.

⁹ 'Science leaders, members of the Society of Polish Youth, scouts, pupils of all schools' (originally 'Przodownicy nauki, zetempowcy, harcerze, uczniowie wszystkich szkół'; authors' own translation from Polish to English).

¹⁰ Since year and month of birth are not reported in these census waves, we cannot construct the measure of exposure to the reform that is required in our difference-in-differences strategy.

Descriptive statis	tics: Analytical	sample for	the a	analysis	of the	1954/55	reform	(Census	data).
Source: 2002 Poli	sh census.								

	Ν	Mean	Standard deviation (for non-dummy variables)	Min	Max
Female	200,706	0.52	-	0	1
Age	200,706	53.91	2.48	50	59
Birth year	200,706	1948	2.48	1943	1952
Secondary educ.	200,706	0.38		0	1
Tertiary educ.	200,706	0.15	-	0	1
Work	198,200	0.47	-	0	1

Note: The table presents descriptive statistics for the analytical sample. The latter includes individuals born between October 1943 and March 1952 in the months of October, November, December, January, February and March.

The main analytical sample includes only individuals born between October 1943 and March 1952. As discussed later, the identification strategy relies on variation that arises from comparing individuals born within a few weeks from the cut-off date to start school every year, which is December 31st. Thus, our primary analytical sample includes only individuals born between October 1st and March 31st of the following calendar year. Individuals born every year between April 1st and September 30th are excluded in order to conduct comparisons between individuals born within a maximum of six months. We examine in Section 7 the robustness of the results to include in the analytical sample individuals born in different months with reassuring results. Furthermore, we exclude individuals born before October 1943 in order to ensure that individuals in our sample were younger than 60 at the time of the census (i.e. spring 2002) and, therefore, not retired.¹¹ Finally, we also exclude individuals born after March 1952 to ensure that individuals in our sample are not affected by subsequent school reforms that occurred during their compulsory education.

Table 1 displays the summary statistics for the analytical sample in the 2002 census. The age of the individuals in the sample ranges between 50 and 59, 38% have completed secondary education, and 15% have completed tertiary education. Only 47% of the individuals in the analytical sample were employed when the survey was implemented. The share of labour force participation among individuals aged between 50 and 59 in the sample is indeed low for both men (53%) and women (41%). While early retirement is unfeasible for men before the age of 60 and very limited for women, the share of active individuals falls at a constant rate from the age of 46 years. This decline in labour force participation for individuals older than 45 is explained by the widespread concessions of disability pensions in the 1990s and before (Rzońca and Wojciechowski, 2008; Czerwiec, 2009). While lack of labour force participation could be partially reflecting some population on disability pensions rather than other forms of inactivity or unemployment, this does not confound our estimates of interest. This is the case because individuals born in the first quarter of the year do not have any explicit benefit regarding access to disability pensions relative to those born in the last quarter of the previous calendar year.

While the main analysis relies on information from the 2002 census, we re-estimate in Appendix A the effects of the reform using the 1998, 1999, and 2000 waves of the Polish Household Budget Survey (HBS).¹² The main drawback of the HBS data is that the sample of individuals born within the period of interest is substantially smaller than in the census data. On the other hand, contrary to the census, the HBS databases include information on household income. Furthermore, we use information on values from three waves of the Diagnoza Społeczna survey conducted in 2011, 2013, and 2015 to explore what mechanisms could be driving the effects of the reform.¹³

4. Empirical strategy

In this section, we describe the difference-in-differences (DiD) strategy used to estimate the effects of the reform implemented in 1954/55 that revoked political indoctrination in school.

The effect of interest is estimated by exploiting the variation in the exposure to the reform over two dimensions. The first one is the different exposure to the reform of individuals born in the last quarter of the year or the first quarter of the following calendar year. In Poland, children start compulsory education in September of the calendar year when they turn seven years old, with the cut-off date thus being December 31st. Therefore, despite being born only a few weeks or months later, individuals born in the first quarter of any calendar year (January to March) begin school nearly one year after those born in the last quarter of the previous calendar year (between October and December). The latter implies that, for those who were already enrolled in compulsory education at the time the reform was put into place, people born in January through March were subject to one more

¹¹ Early retirement before the age of 60 was sparse. This is discussed in detail below in the paper.

¹² The available HBS waves implemented before 1998 and between 2001 and 2010 lack information on the birth date which is necessary to construct the measure of exposure to the reform and conduct our difference-in-differences analysis.

¹³ These were the only waves of the Diagnoza Społeczna survey with information on year and month of birth, which is necessary to construct the variable that measures exposure to the reform and estimate the effects of the reform. We also considered the use of the Polish sample of cross-national surveys that include information on values such as the European Social Survey (ESS), the Generations and Gender Survey (GGS), and the Life in Transition Survey (LIST). However, the lack of information on the month of birth prevented us from using these databases.

year of education under the post-reform educational system than people born in October through December of the preceding year. As explained in the previous section, individuals born between April 1st and September 30th are excluded from he sample in order to be able to conduct meaningful comparisons between individuals born within a maximum of a six-month period. Section 7 reports evidence of the robustness of our estimates to the inclusion of individuals born in different months with reassuring results.

Importantly, those born in the final quarter of the year begin school earlier than children born in the first quarter of the subsequent year. Due to the possibility that earlier school start times will have longer-lasting, more widespread effects on labor and educational outcomes through maturity (see for example Black et al., 2011 and Balestra et al., 2020), the differences between individuals born before and after the cut-off date cannot be exclusively attributed to an additional year of exposure to the post-reform educational system.

To net out the effect of starting school at an older age from the effect of one additional year of education under the postreform educational system, we add a second source of variation in exposure to the reform. We compare the differences between those born in the first quarter of the year and those born in the last quarter of the previous calendar year across two samples of individuals. The first sample includes individuals born between October 1943 and March 1947. These individuals were already enrolled in compulsory education when the reform was implemented in September 1954 and, for them, being born between January and March implies one additional year of education under the reformed education system relative to those born between October and December of the previous calendar year. The second sample includes individuals born between October 1947 and March 1952. These individuals started school after the reform was introduced in September 1954. Therefore, in such a sample, individuals born in the first quarter of the year or the last quarter of the previous calendar year are equally exposed to the reformed education system. In other words, the individuals in the second sample were unexposed to the pre-reform educational system regardless of their month of birth. By comparing the effect of being born in the first quarter of the year relative to being born in the last quarter of the previous calendar year across both samples, we net out maturity effects from the effect of one additional year of education under the reformed educational system. Figs. B.1–B.3 in Appendix B show the number of years of education under the post-reform educational system during compulsory education by year of birth.

Formally, we estimate the following difference-in-differences equation:

$$Outcome_{i,s} = \delta_0 + \delta_1 Q \mathbf{1}_{i,s} \times Inschool_s + \delta_2 Q \mathbf{1}_{i,s} + \gamma_s + X_{i,s} + u_{i,s}$$
(4.1)

where the variable Outcome measures the human capital and labour market outcome of individual i born in half-year s. Because the analytical sample only includes individuals born within three months from December 31st every year (between October and March of the following year), the half-year s indicates the 31st of December of reference. For example, those individuals born between October 1st, 1947 and March 31st, 1948 belong to the same half-year of reference. Equally, individuals born between the 1st of October 1948 and the 31st of March 1949 also belong to the same half-year, and so on. Q1 is a dummy variable equal to 1 if the individual is born between January and March, the first quarter of the year, and 0 if the individual is born between October and December of the previous calendar year. In school is a dummy variable that indicates whether the individual belongs to a half-year that is already in compulsory school when the reform was implemented in September 1954 and 0 if the individual is young enough to start school after the introduction of the education reform examined. Among the group of individuals already in school at the time of the reform, those born in Q1 received one additional year of education after the reform relative to those born in Q4 of the previous calendar year. However, in the group of individuals that enrolled in school after the implementation of the reform, all were equally exposed to the post-reform education system regardless of their exact birth date. γ_s is a vector of half-year fixed-effects and $X_{i,s}$ is a vector of individual-level control variables. The parameter of first interest in the analysis is the interaction term δ_1 , which measures the causal effect of an additional year of exposure to the post-reform educational system with a revoked political indoctrination relative to a year of education in the pre-reform educational system during compulsory education. If indoctrination harms human capital and labour force participation, the effect of the education reform measured by the coefficient δ_1 would have a positive sign.

4.1. Testing identification assumptions

Unlike in standard difference-in-differences designs, the correct identification of the effects of the reform does not rely on the parallel trends condition. Rather, the key assumption in our design is that the pure effect of being born in the first quarter of the calendar year relative to being born in the last quarter of the previous calendar year is the same under the pre-reform and post-reform education systems, in terms of the outcomes explored.

We test the main identification condition for education outcomes by comparing the effect of being born in Q1 relative to being born in Q4 of the previous calendar year across the two samples. The first sample includes only individuals who pursued all their education in the pre-1954 reform educational system (i.e. individuals born between 1926 and 1930).¹⁴ The second sample includes only individuals who started their education after the introduction of the reform (i.e. individuals born between 1947 and 1952). These individuals pursued all their education in the post-reform educational system. This analysis cannot be conducted for labour market outcomes because nearly all individuals born before 1930 were retired from work in 2002 when the data were collected.

¹⁴ The vast majority of them had completed their education by the time the 1954/55 reform was introduced.

Age

lacebo	o effect:	The	effect	of	being	born	in	the	first	quarter	of	the	year	across	different	educational	systems.	
--------	-----------	-----	--------	----	-------	------	----	-----	-------	---------	----	-----	------	--------	-----------	-------------	----------	--

	(1)	(2)	(3)	(4)
			Labour	
	Secondary educ.	Tertiary educ.	force part.	Ln income p/c
Panel A: Educated only before 1954/55	reform vs. only post-1	954/55 reform usin	ng Census	
Q1 \times Only pre-1954 individuals	0.0025	0.0041		
	(0.0045)	(0.0032)		
Q1	0.0074**	0.0013		
	(0.0030)	(0.0022)		
Observations	187,030	187,030		
Cut-off of reference FE	YES	YES		
Sex	YES	YES		
Cut-off of reference FE	YES	YES		
Panel B: Educated only during communism	n vs. only after comm	unism using HBS		
Q1 \times Only educated in communism	-0.0179	-0.0003	-0.0124	-0.0113
	(0.0118)	(0.0116)	(0.0105)	(0.0167)
Q1	0.0150	0.0012	-0.0069	0.0141
	(0.0094)	(0.0104)	(0.0085)	(0.0135)
Observations	26,674	26,674	26,728	26,427
Sex	YES	YES	YES	YES
Cut-off of reference FE	YES	YES	YES	YES
Year of survey FE	YES	YES	YES	YES

Note: The analysis reported in Panel A assesses whether the effect on education of being born in the first quarter of the year relative to being born in the last quarter of the previous calendar year is different in the pre-1954/55 reform educational system and in the post-1954/55 reform educational system. The former includes individuals who completed their education before the 1954/55 reform (aged at least 23 in January 1954) and the latter includes individuals who initiated education after the introduction of the 1954 reform. The dataset used for this analysis is the 2002 Polish census. Effects on labour force participation cannot be estimated since individuals who completed their education before the 1954/55 reform were already retired when the census was implemented in 2002. The analysis reported in Panel B assesses whether the effect of being born in the first quarter of the year relative to being born in the last quarter of the previous calendar year is different for individuals that started education after the 1989/90 post-communism educational reform. The dataset used for this analysis is the HBS rounds 2011–2017. *** p < 0.01, ** p < 0.05, * p < 0.1.

VES

VES

YES

VES

Next, we complement our analysis by checking whether the effect of being born in the first quarter of the year is the same for individuals only exposed to the communist education system, and individuals only exposed to the post-communist education system introduced after the collapse of the Peoples Republic of Poland in 1989 (using HBS data). Specifically, we compare such effect for individuals born between 1982 and 1986 (who started school after the post-communist school reform) and individuals born between 1961 and 1965 (who had completed all their education when the post-communist reform was introduced).

The results of these two analyses are reported in Table 2. The difference in the effect of being born in the first quarter across education systems is identified by the difference-in-differences parameter. The small and statistically insignificant coefficient reported in Panel A suggests that the effect of being born in the first quarter of the year on educational outcomes is very similar for individuals only exposed to the pre-1954 educational system and the post-1954 educational system. Furthermore, the results reported in Panel B reveal that the effect of being born in the first quarter of the year does not seem to be different for individuals only exposed to post-communist education and for those only exposed to communist education. These results reveal that the pure effect of being born in the first quarter of the year does not systems in Poland, suggesting that the main identification condition of our analysis holds.

An additional concern with our analysis is the incorrect assignment of individuals to their first school year. That would increase the measurement error of the Q1 variable and thus bias downwards the estimates of interest. The latter might happen also in the case of grade repetition or late school enrolment. While grade misassignment is uncommon in the 1950s in Poland (Mauersberg, 1996), we lack information on grade repetition. Therefore, our estimates should be interpreted as intention-to-treat estimates and a lower bound for the treatment effect on the treated, which does not threaten the main conclusions of the study.

5. Results

Fig. 1 shows the evolution of human capital outcomes and labour force participation by year of birth for individuals born in Q1 and born in Q4 of the previous calendar year using the census data. Overall, the figure suggests that individuals born in the first quarter of the year have better educational and labour outcomes than those born in the last quarter of the previous calendar year.



Fig. 1. Human capital and labour force participation for individuals born in the first quarter of the year and the last quarter of the previous calendar year. *Note:* The graphs compare the evolution of labour force participation, the share of individuals that completed secondary and the share of individuals that completed tertiary education for those born in the first quarter of the year and for those born in the last quarter of the previous calendar year. The year indicated in the X-axis refers to the year of birth for those born in the first quarter and to the year of birth + 1 for those born in the last quarter of the previous calendar year. Lines represent fits from a local polynomial with a bandwidth of 1 and 95% confidence intervals. *Source:* 2002 Polish census.

The difference between individuals born in these two quarters is however larger for the generations born before 1947, for which being born in the first quarter of the year implies an additional year of exposure to the reform relative to those born in the last quarter of the previous calendar year. This pattern suggests a beneficial (positive) effect of the 1954/55 reform (which removed political indoctrination in schools) on labour force participation and human capital accumulation.¹⁵

The estimated effects of the 1954/55 reform that removed political indoctrination in school are reported in Table 3. The results show that an additional year of education under the reformed educational system during the compulsory education period increases the probability of finishing secondary education, tertiary education, and increases labour force participation later in life relative to a year of education in the pre-reform educational system. Specifically, an additional year of education under the post-1954 education system increases the probability of completing secondary education and university by 1.6 (an increase of 4.2% relative to the sample mean) and 1.3 percentage points (an increase of 8.7%), respectively. The estimated effect of an additional year of education under the post-reform education system on labour force participation in 2002 is 1.5 percentage points (an increase of 3.2%).¹⁶

In Appendix C, we examine the heterogeneous effects of the reform by different dimensions. First, we explore whether the effects of the reform on human capital investments and labour force participation are different for men and women. The results are reported

 $^{^{15}}$ A more detailed graph showing the differences between those born in the first quarter of the year and those born in the last quarter of the previous calendar year by year of birth is reported in Fig. F.1 in Appendix.

¹⁶ A relevant question is whether the increase in labour force participation 50 years after the reform is welfare-enhancing. In principle, labour force participation in older age might not be welfare-enhancing for specific jobs, especially if the alternative is paid retirement. However, it is important to note that all individuals in the analytical sample are younger than 60 and therefore do not qualify for retirement. Furthermore, only a small share of the Polish households surveyed in the Diagnoza Społeczna declared having savings (Czapinski, 2011). In this context, we believe it is unlikely that not working is welfare-enhancing for most blue-collar jobs. Furthermore, evidence has shown that working longer can prevent cognitive decline (Bonsang et al., 2012).

Effect of the 1954/55 reform on education and labour force p	participation	(Census data).
--	---------------	----------------

	1		
	(1)	(2)	(3)
	Labour		
	force part.	Secondary educ.	Tertiary educ.
$Q1 \times In \text{ school}$	0.0142***	0.0169***	0.0115***
	(0.0047)	(0.0047)	(0.0035)
Q1	0.0140***	0.0080***	0.0017
	(0.0031)	(0.0030)	(0.0022)
Observations	198,200	200,706	200,706
Cut-off of reference FE	YES	YES	YES
Sex	YES	YES	YES

Note: This table reports the difference-in-differences estimates of the effects of an additional year of exposure to the post-reform educational system on labour force participation, the probability of completing secondary education, and the probability of completing tertiary education relative to a year of education in the pre-reform educational system during compulsory education. *Source:* The 2002 Polish census. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

in Panel A of Table C.1 and suggest that the effect of the reform on labour force participation is almost identical for men and women. However, we find that the effects of the reform on human capital accumulation are larger for women.¹⁷

Second, we investigate whether the effects of the reform are more pronounced in the western regions of Poland, which witnessed significant forced migration from the Kresy regions following World War II. Recent evidence documents that forced migration heightened preferences for education and increased human capital investments in the descendants of those migrants (Becker et al., 2020). Our findings, as reported in Panel B of Table C.1, reveal that the impact of the reform on labour force participation is approximately two times larger in regions affected by forced migration after WWII, although this difference is not statistically significant. On the other hand, we do not observe any meaningful difference in terms of the reform's impact on education between regions with and without forced migration.

Third, we assess the effects of the reform by grade level. The results of these analyses should be taken with caution as the estimates are notably noisier. Overall, the results reported in Table C.2 show that while the beneficial effect of exposure to an additional year of post-reform education in terms of labour force participation is larger for children in the initial grades, the beneficial effect in terms of the probability of finishing secondary and tertiary education is larger for children in later grades.

6. Mechanisms

This section discusses two potential mechanisms that could be driving the beneficial effects of the reform documented above in the paper.¹⁸ First, we examine whether the effects on human capital accumulation and labour force participation might be driven by the reform affecting a broad set of values and attitudes. Stalinism may have had an impact on values like agency and individualism that ultimately influence judgments about the accumulation of human capital and labor force participation, even though it was not an ideology that was against work or education. For completeness of the analysis, we also examine in Appendix D the effects of the reform on other values, social capital, and political outcomes. Secondly, we test whether the effects might be driven by the reduced instruction time (by three teaching hours per week). In a context where teaching quality might be low, and parents might be more effective in raising offspring human capital, reducing instruction time might improve human capital investments and labour force participation.

While we provide evidence consistent with the hypothesis that the results are driven by an increased individualism and agency after the reform, we cannot rule out other mechanisms, such as teachers' motivation, that we cannot examine empirically.

6.1. Individualism and sense of agency

Previous studies showed that individualism is a crucial driver of economic development (Gorodnichenko and Roland, 2011a, 2012, 2011b; Viinikainen et al., 2010). In this set of studies, individualism is defined as the conviction that people should take care of themselves, as evidenced by the emphasis placed on freedom, independence, self-reliance, competitiveness, and personal action.

 $^{^{17}}$ The differential effects of the reform on education by gender might be influenced by various factors. Firstly, the effect of the reform on values might be different for men and women. While the results should be interpreted with caution due to limited statistical power, we examine the effects of the reform on values for men and women and find limited support for this hypothesis (results are available upon request to the authors). Secondly, the results are consistent with higher returns to education in terms of labour force participation for women. This is documented for post-communist Poland by Myck et al. (2009). However, we lack studies that examine education returns during the communist era in Poland. Further research is necessary to gain a deeper understanding of the diverse effects of the reform on education based on gender.

¹⁸ While we cannot disentangle the effect of removing indoctrination from the educational curriculum from the effects of changes in the grading that accompanied this change (e.g. participation in ideological activities was no longer rewarded with better grades), we believe both components are inseparable parts of any school indoctrination system.

Effect	of	the	1954/55	reform	on	self-direction	and	agency.
--------	----	-----	---------	--------	----	----------------	-----	---------

	(1) Dependence on authorities	(2) Self-reliance	(3) Dependence on others	(4) Dependence on destiny
Q1 \times In school	-0.0500* (0.0271)	0.0969**	0.00290	0.0272
Q1	-0.0496	0.118	-0.145	0.231**
	(0.0735)	(0.0898)	(0.0915)	(0.104)
Mean	0.0783	0.597	0.260	0.419
Observations	1789	1789	1789	1789
Cut-off of reference FE	YES	YES	YES	YES
Year of survey FE	YES	YES	YES	YES
Age	YES	YES	YES	YES
Sex	YES	YES	YES	YES

Note: This table reports the difference-in-differences estimates of the effect of an additional year of exposure to the post-reform educational system on self-direction and agency outcomes relative to a year of education in the pre-reform educational system during compulsory education. All dependent variables are dichotomous variables. *Source:* The analysis uses the 2011, 2013, and 2015 rounds of the Diagnoza Społeczna survey. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

It is plausible that Polish people in the late 1940s and early 1950s, whose predominant and constant emotion could have been fear of being a victim of Stalinist authorities for no specific reason (Diner, 2004), were focused on doing only what they were instructed, and nothing else. The pre-reform school system reflected these values. However, the 1954/55 reform changed the school curriculum and grading enhancing critical thinking, ambition, and self-direction to a greater degree. Children were rewarded for their academic skills rather than for their participation in political activities or obedient behaviour. As a result, the appreciation of one's own independent role in decision-making arguably had more chances to be internalised. These attitudes are relevant for enhanced willingness to put effort into more ambitious educational and professional choices over the entire life course.

To examine the effects of the reform on individualism and self-agency, we rely on three rounds of the Diagnoza Społeczna survey conducted in 2011, 2013, and 2015 which provide information on self-agency and individualism as well as the necessary information to build our measure of exposure to the reform.¹⁹ The survey includes questions about agency and independence from authorities which have been used in previous studies to measure the effect of individualism on development outcomes (Gorodnichenko and Roland, 2011a, 2012, 2011b; Viinikainen et al., 2010).

The results of the effects of the reform are reported in Table 4. Estimates suggest a significant effect of the 1954/55 education reform by increasing the propensity towards personal responsibility and decreasing the perceived influence of authorities. The sense of greater agency and independence from authorities appears to be associated with the ability to shape one's own destiny, which aligns with the observed increases in labour market participation and human capital attainment.

In Appendix D, we exploit additional information on values, attitudes, and other outcomes included in the Diagnoza Społeczna for completeness of the analysis. Specifically, we explore the effects of the reform on beliefs about the most important things in life, the characteristics of the desired job, social capital, and political outcomes. Many of these analyses are estimated with very small sample sizes due to the high number of missing values and the fact that not all the questions are included in each survey wave. Thus, the results should be interpreted with caution and most of the estimates remain small and statistically insignificant. However, taken together, the results suggest that individuals more exposed to the policy are more career-oriented. These results can be interpreted in the light of previous evidence that highlights the importance of individualism and agency for development (Gorodnichenko and Roland, 2011a, 2012, 2011b; Viinikainen et al., 2010).

In the Appendix E, we test further whether the results are driven by indoctrination or other pillars of communist education by examining the effect of the education reform that followed the fall of the communist regime in Poland in 1989. While this reform removed standard features of communist education systems such as the importance of physical education, it expand the teaching of foreign languages and religion, and decentralised school governance. The educational curriculum in the last decade of the Peoples Republic of Poland was *de facto* free of Marxist-Leninist ideology (Mader, 1988). The analysis shows that this reform, which did not affect political indoctrination but rather focused on other aspects of communist education, had a suprisingly limited effects on labour force participation and human capital investments. We interpret this result as additional evidence that political indoctrination in school is the primary driver of the documented effects of the 1954/55 reform on labour force participation and human capital accumulation.

6.2. Reduction in instruction time

While the rest of the curriculum remained the same, removing political indoctrination in school entailed a reduction in the total school instruction time from 29 to 26 h per week in the first seven grades. In this subsection, we examine whether the beneficial

¹⁹ Detailed information on the Diagnoza Społeczna survey is reported in Appendix D.

Effect of the 1974/75 reform that reduced instruction time.									
	(1)	(2)	(3)						
	Labour								
	force part.	Secondary educ.	Tertiary educ.						
Q1 × In school	0.0021	-0.0056	-0.0113***						
	(0.0042)	(0.0045)	(0.0036)						
Q1	-0.0064**	0.0114***	0.0108***						
	(0.0030)	(0.0032)	(0.0026)						
Observations	191,412	196,232	196,232						
Cut-off of reference FE	YES	YES	YES						
Sex	YES	YES	YES						

Note: This table reports the difference-in-difference estimates of the effect of an additional year of exposure to the educational reform implemented in 1974/75 that shortened the instruction time removing the classes in 6 Saturdays per school course relative to a year of education in the pre-reform educational system. *Source:* 2002 Polish census. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

effects of the reform were driven by the reduction in instruction time. While the length of time at school may affect educational outcomes, the broad existing theory points that, if any, the effect of reducing instruction time varies from null to negative. Numerous empirical studies assessing the effects of reforms increasing instruction time (Barrios-Fernandez and Bovini, 2021; Parinduri, 2014; Huebener et al., 2017; Lavy, 2010), reducing instruction time (Kikuchi, 2014), or increasing the length of school year (Pischke, 2007) support the latter conclusion. One may argue however that it is feasible that in contexts that combine long instruction times with poor quality education, a reduction in instruction time could improve school outcomes, particularly if parents are educated enough and substitute poor quality instruction time at school with effective learning at home. This is however unlikely in our setting for different reasons. Firstly, while less beneficial than in developed countries, none of the studies conducted in developing countries or settings with poor quality of education show a negative effect of increasing instruction time (see for example Barrios-Fernandez and Bovini, 2021; Parinduri, 2014; Lavy, 2010). Secondly, the reduction of instruction time from 29 to 26 h per week remains below the average instruction time in Western countries. It is unlikely that the reform increases human capital investment by relieving children from very long and exhausting schooling hours. Finally, in a context in which the level of education of parents is low,²⁰ it is unlikely that school with instruction at home.

We conduct the following analysis to examine further the hypothesis that the effect of the reform could be driven by the reduction in instruction time. To this end, we assess the effect of the introduction of free Saturdays in schools in communist Poland, which resulted in reduced instruction time. While all Saturdays were school days in the early 1970s, six Saturdays were removed from the school calendar starting from the school year of 1974/75. We estimate the effects of this reform using the same empirical approach described in Section 4.²¹

The effects of this analysis are reported in Table 5. The estimates reveal that the reform that reduced the instruction time did not seem to have any significant effect on labour force participation and on the probability of completing secondary education. However, we find a negative effect on the probability of completing tertiary education.

The effects of our empirical exercise, as well as those found in previous studies, suggest that any effects of reducing instruction time are likely to be negative. Therefore, if there is any bias present, the estimates we provided in the previous section for the effects of the 1954/55 reform should be interpreted as a lower bound for the true effect of revoking political indoctrination in the schooling system. This, however, does not alter the main conclusions of the study.

7. Robustness checks

This section summarises the results of a battery of robustness checks, which are reported in detail in Appendix.

First, in the Appendix A we report the evidence testing whether the effect of the 1954/55 reform holds when using the HBS data from rounds 1998, 1999, and 2000 instead of the census data. The results of the analysis using this source of data are reported in Table A.1. Our results support, overall, the conclusions yielded by the main analysis of the paper, showing that an additional year

 $^{^{20}}$ In the census data, only 15.8% and 5.6% of adults equal or older than 25 at the time of the reform in 1954 had secondary studies and tertiary education, respectively.

²¹ To estimate the effect of the 1974/75 reform that reduced annual instruction time by six Saturdays, we compare the difference between those born in the first quarter of the year and those born in the last quarter of the previous calendar year across two samples of individuals. The first sample includes individuals born between October 1963 and March 1967. These individuals were already in compulsory education when the reform was implemented. For these individuals, being born between January and March implies one additional year of education under the education system with fewer days of school relative to those born between October and December of the previous calendar year. The second sample includes individuals born between October 1967 and December 1971. These individuals in this sample are exposed to the same number of years of compulsory education under the post-reform educational system regardless of whether they were born between January and March or between of the previous calendar year across both samples, we net out maturity effects from the first quarter of the year relative to being born in the last quarter of the previous calendar year across both samples, we net out maturity effects from the effect of one additional year of education under the education system with reduced instruction time.

of education under the post-reform educational system increased tertiary education attainment, labour force participation, and also income per capita in later life.

Second, we examine the robustness of our results to alternative definitions of the analytical sample. In the main analysis, we include individuals born within three months of the cut-off date and we exclude those born between April 1st and September 30th. In this section, we test the robustness of the results to the inclusion of individuals born within one month, two months, and four months from December 31st every year. The results of these analyses are reported in Table F.1 of Appendix F. While the magnitude of the effects varies slightly, the results show that the main conclusions are robust to alternative definitions of the analytical sample. We also examine the robustness of the results to removing from the analytical sample those individuals born within one month away from the cut-off date, namely those born in January and December. In other words, we re-estimate the main specification using only individuals born in October, November, February, and March. This analysis would be particularly relevant if those born in December and January are more likely to be assigned to an incorrect school year. The results of this analysis are reported in Table F.2 in Appendix F and show nearly identical results to the main estimates.

Third, we conduct a placebo check falsely setting the reform's introduction date in the school year 1962/63 rather than in 1954/55. The false date is selected so that both falsely exposed and unexposed individuals within the relevant window started school after the 1954/55 reform and were already out of tertiary education by September 1989 when the post-communist education reform was implemented.²² The results of this falsification analysis are reported in Table F.3 in Appendix F showing no effects of the placebo reform on human capital or labour market outcomes.

Fourth, Table F.4 reports the results of a placebo analysis setting as a false cut-off on June 31st rather than on December 31st. This placebo analysis compares individuals born in the second and the third trimester every year. Among this sample of individuals, those born in the second and third trimesters of the year are equally exposed to the reform. Reassuringly, the estimates reported in the table confirm null effects in this placebo check.

Fifth, we re-estimate the results reported in Table F.5 in Appendix F clustering the standard errors at the year of birth level and using wild bootstrapping techniques in order to account for the small number of clusters (Cameron et al., 2008). While more conservative, the estimates remain statistically significant at conventional confidence levels. We also retrieve similar and re-assuring results in Table F.6 when we estimate the main regression including region of residence fixed effects.²³ We address another potential concern, namely whether our estimates are driven by forced migration after the II World War following Becker et al (2020). Table F.7 reports the effects both including and excluding regions exposed to forced migration, and suggest consistent and comparable estimates, which only labour marker participation exhibiting a slightly weaker effect size and significance.

Finally, one potential threat for the interpretation of the coefficient as identifying the effect of the reform on labour force participation is that it might be affected by early retirement. The Polish public pension system was progressively reformed during the 1990s, which made retirement before the age of 65 for men and 60 for women very hard (Góra, 1999; Góra et al., 2008). Before the retirement reform, women aged 55 or above were eligible for early retirement if they had worked for more than 30 years. For men, retirement before the age of 60 (the age threshold in the analytical sample) was not possible, except for agricultural workers. The results reported in Table C.1 show strong labour market effects of the 1954/55 education reform for urban men for whom retirement before the age of 60 was not possible. The latter result suggests that the main results of the study are unlikely driven by differential early retirement among individuals born in the first quarter and the last quarter of the previous calendar year.

8. Conclusion

At its peak, one-third of the world's population lived in communist countries (Eberstadt, 2003). While most of these regimes collapsed in the late 1980s and early 1990s, Marxist-Leninist doctrines continue to inspire various autocratic governments, including China, North Korea, Cuba, Laos, and Vietnam, where approximately 1.6 billion people currently reside.

Although there were numerous differences between education systems in communist and Western Europe (e.g. access to nonpublic schools, length of the school day, homework load, etc.), this paper documents evidence of the long-term effects on labour force participation and human capital accumulation of a fundamental feature of communist education, namely political indoctrination in school. More specifically, we exploit unique evidence from an education reform implemented in mid-1950s Poland, which removed political indoctrination from schools while leaving the rest of the curriculum unchanged. We focus on the effects of repealing the Marxist-Leninist content in the curriculum and evaluation system, which arguably was intended to promote collectivist values, and reward uncritical obedience to ensure the survival of the regime, alongside penalising individual self-direction. Our results provide evidence of the long-lasting adverse effects of political indoctrination in schools on education and labour force participation almost five decades later. While we cannot definitively identify all the mechanisms driving these effects, we find evidence that the reform fostered values related to individual sense of agency. These values have been associated with economic well-being in previous

 $^{^{22}}$ It is not straightforward to set a placebo date in which individuals remain unexposed (or equally exposed) to other education reforms. The school year 1962/63 is the purest placebo date as individuals who initiated school between 1957 and 1968 were not exposed to post-communist education during their educational life.

 $^{^{23}}$ We did not include in the main specification region fixed-effects or a dummy indicating whether the individual lives in an urban area, because this information is only available at the time of the survey in 2002. On the other hand, we lack information on the region of residence at the time of the reform or birthplace, which would have been ideal. Because the region of residence in 2002 might be affected by the reform, including region of residence fixed-effects may create a bad control problem. Nonetheless, the results reported in Table F.6 show that the estimates of interest hardly change when the regression is estimated with region of residence fixed-effects.

Table A.1

Effect of the	1954/55	reform	using	the	HBS	dataset.	
---------------	---------	--------	-------	-----	-----	----------	--

	(1)	(2)	(3) Labour force	(4)		
	Secondary educ.	Tertiary educ.	part.	Ln income p/c		
$Q1 \times In \ school$	0.0135	0.0173*	0.0264**	0.0331**		
	(0.0134)	(0.0089)	(0.0132)	(0.0161)		
Q1	-0.0113	-0.0155**	0.0005	-0.0203*		
	(0.0093)	(0.0060)	(0.0088)	(0.0111)		
Observations	22,203	22,203	22,284	22,091		
Cut-off of reference FE	YES	YES	YES	YES		
Year of survey FE	YES	YES	YES	YES		
Age	YES	YES	YES	YES		

Note: This table reports the difference-in-difference estimates of the effect of an additional year of exposure to the post-reform educational system on labour force participation, income, the probability of completing secondary education and the probability of completing tertiary education relative to a year of education in the pre-reform educational system. *Note*: 1998–2000 rounds of the HBS. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

studies (Slomczynski et al., 1981; Gorodnichenko and Roland, 2011a). Our results are also consistent with earlier studies indicating that the absence of critical and aesthetic discourse in education hinders the instillation of values of self-direction (Torbert, 1978), which in turn is associated with a stronger perception of marketability and a propensity to consider changing jobs (Hall et al., 2018).

A final question of primary interest is whether political indoctrination in non-communist autocratic regimes would yield similar negative effects. While more research is needed to convincingly address this question, the fact that the detrimental effects of Marxist-Leninist indoctrination appear to be linked to the suppression of individual agency may lend credence to the notion that, although autocratic regimes' suppression of critical thought may allow for short-term regime survival, it eventually undermines economic performance over the long run (Diwan and Vartanova, 2020; Dahlum and Knutsen, 2017).

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

Source files (Original data) (Google Drive)

Appendix A. Analysis of the effects of the 1954/55 school reform using the HBS data

In this appendix, we examine the effects of the 1954/55 reform using information from the Polish Household Budget Survey (HSB) instead of the 2002 census. We use in this analysis those rounds of the HBS that include data on month of birth: 1998, 1999, and 2000.²⁴ The main drawback of the HBS data is that the analytical sample in this dataset is substantially smaller than in the census data. On the other hand, the HBS databases include information on household income, which is not available in the Census dataset.

The results of the analysis are reported in Table A.1. The estimates reveal that, relative to a year of education under the pre-reform educational system, an additional year of education under the post-reform educational system increases the probability of finishing tertiary education by 1.7 percentage points, per capita income by 3%, and raises the probability of labour force participation nearly 50 years later by 2.7 percentage points. These findings align with those obtained when estimating the impacts of the 1954/55 reform using the 2002 census data.

On the other hand, unlike the analysis conducted with the census data, we do not observe any statistically significant effect of the reform on the probability of completing secondary education when using the HBS data, although the coefficient shows the expected positive direction. One possible explanation for this divergent result is the notably smaller sample size used in the analysis with the HBS datasets. This reduction in sample size diminishes statistical power, leading to increased standard errors.

Appendix B. Exposure to the post-1954/55 reform educational system by year of birth

The figures in this appendix show the number of years of compulsory education in the post-reform educational system for the individuals included in the main analytical sample by year of birth.

²⁴ The waves of the HBS conducted before 1998 and between 2001 and 2010 lack information on month of birth. Without this information, we cannot construct the measure of exposure to the reform that is used in the difference-in-differences analysis.



Fig. B.1. Year of birth and years of compulsory education in the post-reform educational system. Note: The figure shows the number of years of compulsory education in the post-reform educational system by year of birth.



Fig. B.2. Years of compulsory education in the post-reform educational system for individuals who were already in school when the reform was implemented (born between 1943 and 1947). *Note:* The figure shows the number of years of compulsory education in the post-reform educational system by year and month of birth for individuals who were already in school when the reform was implemented in the school year 1954/55. The number of years of compulsory education both before and after the reform was 8.

Appendix C. Heterogeneous effects of the reform

In this appendix, we examine the heterogeneous effects of the reform by different dimensions. To explore the heterogeneous effects, we expand the main difference-in-difference specification by adding a triple interaction between the variable indicating whether the individual is born in the first quarter of the year (Q1), the variable indicating whether the individual was in school when the reform was implemented (*In School*), and the variable representing the dimension for which we intend to explore the heterogeneous effects (*Heterogeneity Dimension*). Additionally, the specification includes two more interactions: *In School* × *Heterogeneity Dimension* and Q1 × *Heterogeneity Dimension*.

We test for heterogeneous effects of the reform across three dimensions. First, we investigate whether the impacts of the reform on human capital investments and labour force participation are different for men and women. The results of this analysis are reported in Panel A of Table C.1. The results of the analysis suggest that the effect of the reform on labour force participation is

European Economic Review 161 (2024) 104641



Fig. B.3. Years of compulsory education in the post-reform educational system for individuals who started school after the reform was implemented (born between 1947–1952). *Note:* The figure shows the number of years of compulsory education in the post-reform educational system by year and month of birth for individuals who initiated school after the reform was implemented in the school year 1954/55. The number of years of compulsory education both before and after the reform was 8.

nearly identical for men and women. On the other hand, we find that the effects of the reform on human capital accumulation are larger for women.²⁵

Second, we investigate whether the effects of the reform are more pronounced in the western regions of Poland, which witnessed significant forced migration from the Kresy regions following World War II. Recent evidence documents that forced migration heightened preferences for education and increased human capital investments in the descendants of those migrants (Becker et al., 2020). Our findings, as reported in Panel B of Table C.1, reveal that the impact of the reform on labour force participation is approximately two times larger in regions affected by forced migration after WWII, although this difference is not statistically significant. On the other hand, we do not observe any meaningful difference in terms of the reform's impact on education between regions with and without forced migration.

Third, we assess the effects of the reform by grade level. The results of these analyses should be taken with caution as the estimates are notably noisier. Overall, the results reported in Table C.2 show that while the beneficial effect of exposure to an additional year of post-reform education in terms of labour force participation is larger for children in the initial grades, the beneficial effect in terms of the probability of finishing secondary and tertiary education is larger for children in later grades.

Appendix D. Effects of the reform on attitudes, values, and other outcomes

In the main manuscript, we examine the effects of the reform on self-agency and individualism. In this appendix, we assess the impacts of the reform on other values, attitudes, and outcomes. For this, we rely on three rounds of the Diagnoza Społeczna survey conducted over a random sample of individuals aged at least 16 and living in Poland.²⁶ The survey was implemented first in the early 1990s aiming to provide insights into the ongoing socioeconomic changes in Polish society, which required frequent and substantial revisions of the questionnaire. Starting from 2000, the survey was conducted bi-annually collecting data on individual preferences, social capital, and political outcomes, which we use to examine the potential mechanisms through which the effects of the reform on labour force participation and human capital operate. Our analytical sample includes three rounds of the Diagnoza

 $^{^{25}}$ The differential effects of the reform on education by gender might be influenced by various factors. Firstly, the effect of the reform on values might be different for men and women. While the results should be interpreted with caution due to limited statistical power, we examine the effects of the reform on values for men and women and find limited support for this hypothesis (results are available upon request to the authors). Secondly, the results are consistent with higher returns to education in terms of labour force participation for women. This is documented for post-communist Poland by Myck et al. (2009). However, we lack studies that examine education returns during the communist era in Poland. Further research is necessary to gain a deeper understanding of the diverse effects of the reform on education based on gender.

²⁶ More information about the Diagnoza Społeczna can be found at http://www.diagnoza.com/index-en.html.

Table C.1

Heterogeneous effects of the	1954–55	reform 1	by gender	and by	whether	the	individual	lives	in a	a region	that
experienced forced migration	after Wo	rld War	II (2002 C	Census):							

Panel A: Results by gender				
	(1)	(2)	(3)	(4)
	Labour			Labour force part.
	force part.	Secondary educ.	Tertiary educ.	(Urban only)
Q1 \times In school \times Female	0.0002	0.0166*	0.0147**	0.0022
	(0.0090)	(0.0090)	(0.0067)	(0.0110)
Q1 × In school	0.0127*	0.0071	0.0034	0.0112
	(0.0069)	(0.0065)	(0.0048)	(0.0085)
Observations	198,200	200,706	200,706	133,252
Cut-off of reference FE	YES	YES	YES	YES
Panel B: Results by living i	n a region with force	d migration		
		(1)	(2)	(3)
		Labour		
		force part.	Secondary educ.	Tertiary educ.
$Q1 \times In \text{ school} \times Forced m$	igration region	0.0083	0.0013	-0.0018
		(0.0091)	(0.0090)	(0.0068)
$Q1 \times In \text{ school}$		0.0081	0.0155**	0.0122**
		(0.0070)	(0.0068)	(0.0052)
Observations		198,200	200,706	200,706
Cut-off of reference FE		YES	YES	YES

Note: This table reports the heterogeneous effects of the reform on labour force participation, the probability of completing secondary education, and the probability of completing tertiary education relative to a year of education in the pre-reform educational system during compulsory education. Panel A reports the heterogeneous effects by gender and Panel B reports the heterogeneous effects by whether the individual lives in a region affected by forced migration after II World War. To examine the heterogeneous effects, we expand the baseline difference-in-differences specification by adding a triple interaction between $Q1 \times In$ school × Female in Panel A and $Q1 \times In$ school × Forced migration regions in Panel B. Female and Forced migration regions are dummy variables equal to 1 if the individual is a woman and if the individual lives in a region with forced migration. We also include $Q1 \times$ Female and In school × Female in Panel A and $Q1 \times$ Forced migration in Panel B. Source: 2002 Polish census. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Table C.2

Effect of the 1954/55 reform by grade.

	(1)	(2)	(3)
	Labour		
	force part.	Secondary educ.	Tertiary educ.
$Q1 \times 4$ th grade	0.0108	0.0261***	0.0199***
	(0.008)	(0.008)	(0.006)
$Q1 \times 3rd$ grade	0.0112	0.0256***	0.0204***
	(0.008)	(0.008)	(0.006)
Q1 \times 2nd grade	0.0169**	0.0232***	0.0103*
	(0.008)	(0.008)	(0.006)
$Q1 \times 1st$ grade	0.0164**	-0.0008	0.0003
	(0.007)	(0.007)	(0.005)
Q1	0.0140***	0.0080***	0.0017
	(0.003)	(0.003)	(0.002)
Observations	198,200	200,706	200,706
Cut-off of reference FE	YES	YES	YES
Sex	YES	YES	YES

Note: This table reports the heterogeneous effects of an additional year of exposure to the post-reform educational system relative to a year of education in the pre-reform educational system by grade. The outcome variables examined include labour force participation, the probability of completing secondary education, and the probability of completing tertiary education. *Source:* 2002 Polish census. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Społeczna survey, implemented in 2011, 2013, and 2015. These were the only rounds of the survey that include information on the year and month of birth of the interviewees, which is essential information to construct our measure of exposure to the reform.²⁷

²⁷ The survey round conducted in 1995 includes also information on year and month of birth. However, the 1995 round differs with respect to the design of the questionnaire and hardly includes any comparable question. Therefore, we only use in our analysis the data from the 2011, 2013, and 2015 rounds of the Diagnoza Społeczna Survey, which provide harmonised information for a large set of individual values.

Table D 1

Effec	t of	the	1954/55	reform	on	beliefs	about	important	things	in	lif

iffect of the 1954/55 reform on beliefs about important things in life.									
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
	Work	Money	Health	Integrity	Respect	Children	Marriage	Friends	Optimism
Q1 \times In school	0.0380	0.0221	-0.0172	0.0465	0.0129	0.0639	0.0213	-0.0175	-0.0136
	(0.0358)	(0.0410)	(0.0422)	(0.0306)	(0.0266)	(0.0485)	(0.0490)	(0.0199)	(0.0283)
Q1	0.301*	0.502***	-0.231	-0.0104	-0.162*	0.0102	0.0464	-0.0750	-0.147
	(0.166)	(0.0695)	(0.185)	(0.0247)	(0.0899)	(0.198)	(0.198)	(0.0674)	(0.0897)
Observations	1791	1791	1791	1791	1791	1791	1791	1791	1791
Cut-off of reference FE	YES	YES	YES	YES	YES	YES	YES	YES	YES
Year of survey FE	YES	YES	YES	YES	YES	YES	YES	YES	YES
Age	YES	YES	YES	YES	YES	YES	YES	YES	YES
Sex	YES	YES	YES	YES	YES	YES	YES	YES	YES

Note: This table reports the difference-in-differences estimates of the effect of an additional year of exposure to the post-reform educational system on beliefs about important things in life relative to a year of education in the pre-reform educational system during compulsory education. All dependent variables are dichotomous variables. Source: The analysis uses the 2011, 2013, and 2015 rounds of the Diagnoza Społeczna survey. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Table D.2

Effect of the 1954/55 reform on desired job characteristics.

	(1) Social esteem	(2) Independence	(3) Personal development	(4) Professional promotion	(5) Stress free	(6) Employment stability	(7) Convenient hours	(8) Proper wage
Q1 \times In school	0.205***	0.0847	-0.0350	0.0131	-0.0575	-0.0228	0.0129	-0.0903
	(0.0690)	(0.121)	(0.0806)	(0.0146)	(0.129)	(0.136)	(0.0902)	(0.132)
Q1	-0.0714**	-0.233***	0.0115	-0.0224	0.242***	-0.266***	-0.135***	0.176***
	(0.0345)	(0.0593)	(0.0303)	(0.0181)	(0.0602)	(0.0610)	(0.0457)	(0.0582)
Observations	360	360	360	360	360	360	360	360
Cut-off of reference FE	YES	YES	YES	YES	YES	YES	YES	YES
Year of survey FE	YES	YES	YES	YES	YES	YES	YES	YES
Age	YES	YES	YES	YES	YES	YES	YES	YES
Sex	YES	YES	YES	YES	YES	YES	YES	YES

Note: This table reports the difference-in-differences estimates of the effect of an additional year of exposure to the post-reform educational system on desired job characteristics relative to a year of education in the pre-reform educational system during compulsory education. All dependent variables are dichotomous variables. Source: The analysis uses the 2013 and 2015 rounds of the Diagnoza Społeczna survey. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

We examine the effects of the reform on four different categories of outcomes: beliefs about more important things in life, job preferences, political attitudes and preferences, and social capital outcomes. The results of the Diagnoza Społeczna survey should however be interpreted as suggestive due to limited statistical power. While the surveys are nationally representative, we only include in the analytical sample individuals born between October 1943 and March 1952. Furthermore, not all outcomes are collected in every survey. The sample size is even smaller for some outcomes that were only collected in one round of the survey or for a specific subsample of the general population.

We start examining the effects of the reform on beliefs about what things are the most important in life. The results of this analysis are reported in Table D.1. All the estimates are statistically indistinguishable from zero although, taken together, the direction of the coefficients suggests that the reform could have made individuals more career-oriented (work, money). The effects of the reform on the value of relations are less clear, with coefficients changing signs for the importance of friends, children, or marriage.

Table D.2 reports the estimates of the effects of the reform on desired job characteristics for the subsample of individuals who were working at the time of the interview. Therefore, although the analytical sample comprises two waves, the sample is particularly small in these analyses. Nonetheless, the results show a positive and statistically significant effect of the reform on the perception that individuals hold jobs that are socially esteemed or highly respected. This can be treated as an indication that a professional career is more likely to fulfil individual ambitions. The rest of the coefficients are much smaller and statistically indistinguishable from zero at conventional confidence levels.

The estimates reported in Table D.3 show the effects of the reform on political outcomes. The results suggest a positive effect of the reform that reduced indoctrination on the probability of voting for the left-wing party SLD and no effects on support for left-wing views. While the former result could be consistent with a backlash effect of communist indoctrination on the probability of voting for left-wing parties (Sieklucki, 2006), it is important to consider that the effects on these outcomes are estimated using very small and selected sample sizes.²⁸ We also explore the effects of the reform on views of the transition process that transformed the Polish People's Republic into a liberal democracy. While the coefficient of interest is negative, indicating that individuals exposed to reform are more critical of the regime transformation process, the estimates are statistically indistinguishable from zero at conventional confidence levels.

²⁸ For example, only a small fraction of participants responded to the voting question.

Table D.3

Effect of the 1954/55 reform on political outcomes.

	(1) SLD party (left-wing) voter	(2) Left-wing views	(3) Successful transformation
$Q1 \times In \text{ school}$	0.211***	0.359	-0.0325
	(0.0774)	(1.040)	(0.0327)
Q1	-0.0906*	0.870	-0.0605**
	(0.0513)	(0.723)	(0.0281)
Observations	123	106	1417
Cut-off of reference FE FE	YES	YES	YES
Year of survey FE	YES	YES	YES
Age	YES	YES	YES
Sex	YES	YES	YES

Note: This table reports the difference-in-differences estimates of the effect of an additional year of exposure to the post-reform educational system on political outcomes relative to a year of education in the pre-reform educational system during compulsory education. Three outcome variables are examined: the probability of declaring voting for the left-wing SLD Party, the probability of declaring having left-wing views, and the probability of believing the transition process that followed the fall of the communist regime was a successful process. *SLD party voter* and *Successful transformation* are dichotomous variables. The values of the variable *Left-wing views* range between 0 (extreme right-wings views) and 10 (extreme left-wings views). *Source:* The analysis uses the 2011, 2013, and 2015 rounds of the Diagnoza Społeczna survey. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Table D.4

Effect of the 1954/55 reform on social capital (membership of different organisations).

	(1) Sport club	(2) Professional association	(3) Political party	(4) Charity	(5) Trade union	(6) Hobby	(7) Dwelling community	(8) Ecological organisation	(9) Social club	(10) Religious organisation	(11) Knowledge popularisation	(12) Local government	(13) Other
Q1 \times In school	0.124*	-0.0721	0.161*	0.120	0.104	-0.169	0.138*	-0.000393	-0.0257	0.0807	-0.0152	0.141	-0.102
	(0.0745)	(0.0492)	(0.0867)	(0.0767)	(0.0992)	(0.121)	(0.0819)	(0.0745)	(0.155)	(0.187)	(0.153)	(0.0947)	(0.147)
Q1	-0.0512	0.0604	-0.0884	-0.123**	-0.0665	0.0936	-0.125**	-0.000330	0.0397	-0.0833	0.100	-0.0196	0.0129
	(0.0451)	(0.0460)	(0.0671)	(0.0601)	(0.0896)	(0.0852)	(0.0590)	(0.0659)	(0.0995)	(0.117)	(0.0857)	(0.0617)	(0.105)
Observations	118	118	118	118	119	119	119	118	120	121	118	119	120
Cut-off of reference FE	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Year of survey FE	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Age	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Sex	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES	YES

Note: This table reports the difference-in-differences estimates of the effect of an additional year of exposure to the post-reform educational system on membership of different organisations relative to a year of education in the pre-reform educational system during compulsory education. All dependent variables are dichotomous variables. Source: The analysis uses the 2015 round of the Diagnoza Społeczna survey. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Finally, we use information on participation in different organisations collected in the 2015 wave to examine the effects of the reform on social capital. The results are presented in Table D.4. The estimates show that the reform increased the probability of participating in a neighbourhood organisation, political party, and sports club. On the other hand, we do not find significant effects of the reform on participation in other social organisations. The results are consistent with previous evidence that suggests exposure to communism and STASI's mass surveillance decreased social capital and generalised trust (Lichter et al., 2020; Costa-Font and Nicińska, 2023).

Appendix E. The impact of the post-communist education reform (1989/90)

This appendix examines the effects of the education reform that followed the free and democratic parliament elections that took place in June 1989 after the collapse of the communist regime in Poland. The reform, implemented starting in the 1989/90 school year, decentralised schooling, provided alternatives to public education (including religious and private schools), and eliminated Russian as a compulsory foreign language, as well as communist civic education, from the curricula (Lopez and Marlow-Ferguson, 2002; Szczepanek, 2018; Szebenyi, 1992). The reformed education system prioritised the teaching of foreign languages (especially Western European languages), introduced a new program in civic education (Janowski, 1999), and granted teachers substantial freedom in the choice of curricula contents, allowing for the development of innovative educational programmes (Szczepanek, 2018).

However, already in 1981, the growing power of *Solidarność*,²⁹ an anti-communist social movement, affected the school curricula (Mader, 1988) and increased security of teachers' employment. Many teachers supported or joined *Solidarność* (Bochwic, 2000). The conflict between communist authorities and school pupils supported by their parents was open in the 1980s (Janowski,

²⁹ Solidarność was the largest independent trade union in Poland. Founded in 1980, the trade union growed rapidly into a social movement gathering ten million members. It was instrumental in weakening the communist regime in Poland, leading to free parliamentary elections in 1989 and in consequence to the collapse of communism (Goddeeris, 2008).



Fig. E.1. Human capital and labour market outcomes for individuals born in the first quarter of the year and the last quarter of the previous year. *Note:* The graphs are built using data from rounds 2011–2017 of the HBS survey. They compare labour force participation, probability of completing tertiary and secondary education, and income for individuals born in the first quarter of the year and those born in the last quarter of the previous calendar year. The year indicated in the *X*-axis refers to the year of birth for those born in the first quarter or to the year of birth +1 for those born in the last quarter of the previous calendar year. Lines represent fits from a local polynomial with a bandwidth of 1.

Descriptive statistics	Descriptive statistics: Analytical sample for the analysis of the 1989/90 reform (HBS dataset).									
	Ν	Mean	Standard deviation	Min	Max					
			(for non-dummy variables)							
Female	40,455	0.51	_	0	1					
Age	40,455	34.21	2.90	30	42					
Birth year	40,455	1980	2.87	1976	1988					
Secondary educ.	40,347	0.68	-	0	1					
Tertiary educ.	40,347	0.37	-	0	1					
Work	40,455	0.80	-	0	1					
Income p/c	40,100	1456.81	1102.84	0	25,000					

 Table E.1

 Descriptive statistics: Analytical sample for the analysis of the 1989/90 reform (HBS dataset)

Source: The table presents descriptive statistics for the analytical sample. The latter includes individuals born between October 1976 and March 1988 in the months of October, November, December, January, February, and March. *Source:* HBS waves 2011–2017.

1992). The last time the Minister of Education referred to school communities in the inauguration speech as 'comrades'³⁰ was in 1979 (Wagner, 2018). Abucewicz (2009) argues that the increasing role of liberal values in the Polish education system and its management was present throughout the 1980s. The late 1980s in Poland were relatively free from indoctrination (Janowski, 1999).³¹ The courses aimed at instilling communist ideology "were almost unanimously made light of by all concerned, that

³⁰ Originally 'towarzysze'; authors' own translation from Polish to English.

³¹ In spite of the authorities' efforts to intensify communist indoctrination after the end of martial law in 1983 (Wagner, 2018).

Table E.2

Effect of the post-communist education system in Poland (HBS dataset).

-				
	(1)	(2)	(3)	(4)
	Secondary educ.	Tertiary educ.	Labour force part.	Ln income p/c
Q1 \times In school	0.0081	0.0069	0.0147	-0.0004
	(0.0106)	(0.0115)	(0.0094)	(0.0152)
Q1	0.0185**	0.0077	-0.0050	0.0103
	(0.0093)	(0.0103)	(0.0084)	(0.0134)
Observations	40,347	40,347	40,455	40,100
Cut-off of reference FE	YES	YES	YES	YES
Year of survey FE	YES	YES	YES	YES
Age	YES	YES	YES	YES
Sex	YES	YES	YES	YES

Note: This table reports the effects of an additional year of exposure to the post-communist educational system on labour force participation, per capita income, the probability of completing secondary education, and the probability of completing tertiary education relative to a year of education in the communist education system during compulsory education. *Source:* HBS waves 2011–2017. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Table E.3

Effect of an additional year of exposure to post-communist post-compulsory education.

	(1)	(2)	(3)	(4)
	Secondary educ.	Tertiary educ.	Labour force part.	Ln income p/c
Q1 \times In school	0.0295***	0.0098	-0.0061	0.0156
	(0.0106)	(0.0086)	(0.0087)	(0.0141)
Q1	-0.0015	0.0040	0.0078	0.00670
	(0.0077)	(0.0057)	(0.0067)	(0.0104)
Observations	35,082	35,082	35,146	34,689
Cut-off of reference FE	YES	YES	YES	YES
Sex	YES	YES	YES	YES
Year of survey FE	YES	YES	YES	YES
Age	YES	YES	YES	YES

Note: This table reports the effects of an additional year of exposure to the post-communist educational system on labour force participation, per capita income, the probability of completing secondary education, and the probability of completing tertiary education relative to a year of education in the communist education system during post-compulsory education. *Source:* HBS 2011–2017. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

is, students, teachers, and parents. These subjects, introduced in the late 1960s, received no respect and were considered to be unnecessary and unimportant" as Janowski (1999) reports. In sum, the education reform of 1989/90 had little impact on the indoctrination at school in Poland.

For the main analysis of this reform, we rely on seven rounds of the Polish Household Budget Survey (HBS) that collected yearly information between 2011 and 2017.³² We employ the same empirical strategy used in the main analysis of the paper comparing individuals born in the first and in the last quarter of the previous calendar year across two samples. The first sample includes individuals born between 1976 and 1982. These individuals were in compulsory school at the time of the 1989/90 reform and, therefore, were exposed at least to one year of education under the pre-reform communist education system. For these individuals, those born in the first quarter of the year were exposed to one additional year of post-communist education relative to those born in the last quarter of the previous calendar year. The second sample includes individuals born between 1982 and 1988. These individuals started school after the collapse of the communist regime in Poland and, therefore, their level of exposure to the reform is the same regardless of whether they were born in the first quarter of the year or in the last quarter of the previous calendar year. In order to reduce the potential noise caused by a later entrance into the labour market of individuals who invest more in their human capital, we restrict the analysis to individuals who were at least 30 by the time of the survey. The 2002 Census data cannot be used to examine the effect of the reform because most of the individuals in relevant cohorts were too young to even have finished tertiary education by 2002.

Table E.1 provides summary statistics for the analytical sample used in the analysis of the 1989/90 reform conducted using the HBS data. The age of the individuals in the sample ranges between 30 and 42 at the time of the survey, and 37% of them have completed tertiary education. 80% of the individuals in the analytical sample have a job, and 68% completed secondary education. The mean monthly per-capita income is 1457 zloties (approximately 413 USD).

Fig. E.1 shows the evolution of human capital outcomes and labour force participation by year of birth in the past for individuals born in Q1 and in Q4 of the previous calendar year using the HBS data. Most individuals born before 1966 should

 $^{^{32}}$ HBS is a household budget survey collecting data over a representative sample of Polish households on household incomes, expenditures, and living conditions, as well as selected socio-economic characteristics of individual household members. Although the survey dates back to the 1950s, its design has undergone substantial changes in the 1990s and 2000s. Rounds 2003–2010 do not include information on month of birth and therefore cannot be used in the analysis of the effects of the reform.

Table F.1

Effect of the 1954-55 reform using different time windows to select the analytical sample (Census data).

	(1)	(2)	(3)
	Labour force part.	Secondary educ.	Tertiary educ.
Panel A: 1 month window			
First month of year \times In school	0.0193**	0.0282***	0.0079
2	(0.0083)	(0.0082)	(0.0062)
First month of year	0.0131**	0.0092*	0.0105***
·	(0.0054)	(0.0052)	(0.0039)
Observations	65,917	66,724	66,724
Cut-off of reference FE	YES	YES	YES
Sex	YES	YES	YES
Panel B: 2 months window			
First two months of year \times In school	0.0126**	0.0197***	0.0126***
	(0.0058)	(0.0058)	(0.0043)
First two months of year	0.0150***	0.0091**	0.0042
	(0.0038)	(0.0037)	(0.0028)
Observations	129,159	130,826	130,826
Cut-off of reference FE	YES	YES	YES
Sex	YES	YES	YES
Panel C: 4 months window			
First four months of year \times In school	0.0146***	0.0139***	0.0100***
	(0.0040)	(0.0040)	(0.0030)
First four months of year	0.0052*	0.0037	-0.0015
	(0.0026)	(0.0026)	(0.0019)
Observations	265,677	268,979	268,979
Cut-off of reference FE	YES	YES	YES
Sex	YES	YES	YES

Note: This table reports the difference-in-differences estimates of the effects of an additional year of exposure to the post-reform educational system on labour force participation, the probability of completing secondary education, and the probability of completing tertiary education relative to a year of education in the pre-reform educational system during compulsory education. Each panel uses a different sample in which the analytical sample is defined by whether the individual is born within one, two, or four months from the cut-off date each year (December 31st). *Source:* The 2002 Polish census. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Table F.2

Effect of the 1954–55 reform excluding from the analytical sample individuals born in December and January every year.

	(1) Labour force part.	(2) Secondary educ.	(3) Tertiary educ.
$Q1 \times In \text{ school}$	0.013**	0.011**	0.013***
	(0.006)	(0.006)	(0.004)
Q1	0.014***	0.007**	-0.003
	(0.004)	(0.004)	(0.003)
Observations	132,283	133,982	133,982
Cut-off of reference FE	YES	YES	YES
Sex	YES	YES	YES

Note: This table reports the difference-in-differences estimates of the effects of an additional year of exposure to the post-reform educational system on labour force participation, the probability of completing secondary education, and the probability of completing tertiary education relative to a year of education in the pre-reform educational system during compulsory education. The analytical sample uses individuals born within three months from the cut-off every year (December 31st) but excluding those born in January and December. These are arguably individuals with a higher probability of misassignment of the degree of exposure to the reform. *Source:* The 2002 Polish census. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1

have completed their compulsory schooling before the introduction of the post-communist education system in 1989/90. If they undertook post-compulsory education, individuals born between 1966 and 1975 were exposed to at least one year of education under the post-communist education system. Individuals born between 1976 and 1982 were exposed to at least a year of post-communist education during their compulsory schooling. Finally, individuals born after 1982 started school after the introduction of the post-communist education system.

The figure suggests that, although the educational attainment and labour market outcomes have in general improved over time, these outcomes have evolved very similarly for individuals born in Q1 or in Q4 of the previous calendar year since the 1960s. The latter indicates that the post-communist education reform had no large effects on labour market outcomes or human capital investments.

Table F.3

1962/63.	ın	reform	placebo	а	of	Effects
(1)						

	(1) (2)		(3)
	Labour force part.	Secondary educ.	Tertiary educ.
Q1 × In school	0.0029	-0.0050	-0.0021
	(0.0039)	(0.0040)	(0.0030)
Q1	0.0042	0.0058**	0.0016
	(0.0027)	(0.0028)	(0.0022)
Observations	257,632	261,729	261,729
Cut-off of reference FE	YES	YES	YES
Sex	YES	YES	YES

Note: This table reports the difference-in-differences estimates of the effects of an additional year of exposure to a false reform in 1962. *Source:* The 2002 Polish census. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Table F.4

Placebo analysis comparing individuals born in the second and third quarter of the same calendar year.

	(1)	(2)	(3)	
	Labour force part.	Secondary educ.	Tertiary educ.	
$Q2 \times In \ school$	-0.004	0.005	0.003	
	(0.004)	(0.004)	(0.003)	
Q2	-0.006**	-0.003	-0.003	
	(0.003)	(0.003)	(0.002)	
Observations	197,081	199,517	199,517	
Cut-off of reference FE	YES	YES	YES	
Sex	YES	YES	YES	

Note: This table reports the difference-in-difference estimates of the effect of a placebo analysis assessing the effect of being born in the second quarter of the year relative to those born in the third quarter of the same year for cohorts in school at the time of the reform and not yet in school at the time of the reform. Individuals born in Q2 and Q3 within the same calendar year are equally exposed to the reform. *Source:* 2002 Polish census. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Table F.5

Effect of the 1954–55 reform that removed ideology from educational curriculum (2002 Census): Wild bootstrapped standard errors clustered at the running variable level.

	(1)	(2)	(3)
	Labour force part.	Secondary educ.	Tertiary educ.
$Q1 \times In \ school$	0.0142***	0.0169***	0.0115**
	(0.003)	(0.005)	(0.004)
	[0.013]	[0.099]	[0.068]
Observations	198,200	200,706	200,706
Cut-off of reference FE	YES	YES	YES
Sex	YES	YES	YES

Note: This table reports the difference-in-differences estimates of the effect of an additional year of exposure to the post-reform educational system on labour force participation, the probability of completing secondary education, and the probability of completing tertiary education relative to a year of education in the pre-reform educational system during compulsory education. *Note:* 2002 Polish census. Wild-bootstrapped standard errors based on 999 replications and Rademacher weights are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

The estimated effects of the 1989/90 education reform are reported in Table E.2 and reveal that the effect of a higher degree of exposure to the post-communist education system has statistically insignificant effects on human capital investments and labour market outcomes. While the coefficients measuring the effect of an additional year of exposure to post-communist education in Poland on the probability of completing secondary and tertiary education are small, the magnitude of the coefficient measuring the effect on labour force participation is sizeable, although statistically insignificant. Therefore, the latter results should be interpreted with caution, as the absence of statistical significance could be a result of limited statistical power due to the smaller sample size in the HBS database.

The lack of effect of the post-communist school reform in Poland contrasts with the beneficial effects of exposure to postcommunist education in East Germany found in Fuchs-Schündeln and Masella (2016). One potential explanation for the different results could be that we are focusing on compulsory years of education (primary education and the initial years of secondary education), while Fuchs-Schündeln and Masella (2016) concentrate on secondary education. To make results more comparable, we also estimate the effect of exposure to an additional year of non-compulsory secondary education in the post-communist education system on human capital and labour market outcomes. In this analysis, we only include individuals who were old enough to have

Table F.6

Effect of the 1954/55 reform on educ	tion and labour force participation	(Census data): With region of residence
fixed effects.		

	(1) Labour force part.	(2) Secondary educ.	(3) Tertiary educ.
$Q1 \times In \text{ school}$	0.0119***	0.0157**	0.0106**
	(0.0030)	(0.0050)	(0.0038)
Q1	0.0120***	0.0075***	0.0013
	(0.0027)	(0.0012)	(0.0008)
Observations	198,200	200,706	200,706
Cut-off of reference FE	YES	YES	YES
Sex	YES	YES	YES
Region of residence FE	YES	YES	YES

Note: This table reports the difference-in-differences estimates of the effects of an additional year of exposure to the post-reform educational system on labour force participation, the probability of completing secondary education and the probability of completing tertiary education relative to a year of education in the pre-reform educational system during compulsory education. The specification includes region of residence fixed effects. Source: The 2002 Polish census. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

Table F.7

Effect of the 1954/55 reform (using Census data): regions with and without forced migration.

	(1) Labour force part.	(2) Labour force part.	(3) Secondary educ.	(4) Secondary educ.	(5) Tertiary educ.	(6) Tertiary educ.
$Q1 \times In \text{ school}$	0.0151**	0.0086	0.0197***	0.0130*	0.0116**	0.0113**
	(0.0062)	(0.0071)	(0.0063)	(0.0070)	(0.0046)	(0.0053)
Q1	0.0113***	0.0132***	0.0074*	0.0073	0.0040	-0.0024
	(0.0041)	(0.0046)	(0.0040)	(0.0045)	(0.0029)	(0.0034)
Observations	108,073	90,127	109,272	91,434	109,272	91,434
Cut-off of reference FE	YES	YES	YES	YES	YES	YES
Sex	YES	YES	YES	YES	YES	YES
Sample: Regions with forced migration	YES	NO	YES	NO	YES	NO

Note: This table reports the difference-in-difference estimates of the effect of an additional year of exposure to the post-reform educational system on labour force participation, the probability of completing secondary education, and the probability of completing tertiary education relative to a year of education in the pre-reform educational system during compulsory education. The effects are estimated separately for regions with forced migration after World War II as described in Becker et al. (2020) and without forced migration. *Note:* 2002 Polish census. Robust standard errors are reported in parentheses. *** p < 0.01, ** p < 0.05, * p < 0.1.

completed tertiary education before the 1989/90 reform (those born between 1962–1965) and individuals still in post-compulsory education when the 1989/90 reform was implemented (individuals born between 1972 and 1975). The results of this analysis are reported in Table E.3. The estimates of the effects of exposure to an additional year of post-communist education during non-compulsory secondary school are small and statistically indistinguishable from zero at conventional confidence levels except for secondary education.³³

These results reassure the hypothesis that, unlike in East Germany, the education system that was in place in the 1980s in Poland did not have a strong negative effect in terms of human capital investments and labour market outcomes relative to post-communist education. We interpret these results in a context in which ideological indoctrinating contents in the Polish school curriculum in the 1980s were very limited.

Appendix F. Additional tables and graphs

See Fig. F.1 and Tables F.1-F.7

Appendix G. Supplementary data

Supplementary material related to this article can be found online at https://doi.org/10.1016/j.euroecorev.2023.104641.

³³ To make estimates more comparable, we use individuals aged at least 24 when the reform was introduced. Most of these individuals were too old to be attending secondary or tertiary education when the reform was introduced, and as a result, their level of exposure to the reform is not influenced by their quarter of birth.



Fig. F.1. Effect of being born in Q1 relative to being born in Q4 of the previous calendar year by year of birth. *Note:* Figure reports the estimates of the effect of being born in Q1 relative to being born in Q4 of the previous year by year of birth. For those born between 1947 and 1951, all individuals are equally exposed to the reform regardless of their month of birth. For those born between 1943 and 1947, those born in Q1 have an additional year of exposure to the post-reform education system relative to those born in Q4 of the previous year. Data used is the 2002 Polish census.

References

Abadie, A., Diamond, A., Hainmueller, J., 2015. Comparative politics and the synthetic control method. Am. J. Polit. Sci. 59 (2), 495-510.

Abucewicz, M., 2009. System edukacji jako czynnik sprawowania kontroli społecznej i dominacji ideologicznej. Prace Inst. Profil. Społecznej Resocjal. 14, 33–53.
Acemoglu, D., Fergusson, L., Robinson, J., Romero, D., Vargas, J.F., 2020. The perils of high-powered incentives: Evidence from Colombia's false positives. Am. Econ. J.: Econ. Policy 12 (3), 1–43.

Acemoglu, D., Robinson, J.A., 2005. Economic Origins of Dictatorship and Democracy. Cambridge University Press.

- Alesina, A., Fuchs-Schündeln, N., 2007. Goodbye Lenin (or not?): The effect of communism on People's preferences. Am. Econ. Rev. 97 (4), 1507-1528.
- Anderson, J., 1994. Religion, State and Politics in the Soviet Union and Successor States. Cambridge University Press.

Arendt, H., 1951. The Origins of Totalitarianism. Harcourt, Brace and Co, New York.

Arold, B., 2022. Evolution vs. Creationism in the Classroom: The Lasting Effects of Science Education. Technical Report 379, ifo Working Papers.

Aspachs-Bracons, O., Clots-Figueras, I., Costa-Font, J., Masella, P., 2008. Compulsory language educational policies and identity formation. J. European Econ. Assoc. 6 (2-3), 434–444.

Balestra, S., Eugster, B., Liebert, H., 2020. Summer-born struggle: The effect of school starting age on health, education, and work. Health Econ..

Barrios-Fernandez, A., Bovini, G., 2021. It's time to learn: School institutions and returns to instruction time. Econ. Educ. Rev. 80.

Becker, S.O., Grosfeld, I., Grosjean, P., Voigtländer, N., Zhuravskaya, E., 2020. Forced migration and human capital: Evidence from post-WWII population transfers. Am. Econ. Rev. 110 (5), 1430–1463.

Benavot, A., 1992. Curricular content, educational expansion, and economic growth. Comp. Educ. Rev. 36 (2), 150-174.

Bernheim, B., Garrett, D.M., Maki, D.M., 2001. Education and saving: The long-term effects of high school financial curriculum mandates. J. Public Econ. 80 (3), 435–465.

Black, S.E., Devereux, P.J., Salvanes, K.G., 2011. Too Young to leave the nest? The effects of school starting age. Rev. Econ. Stat. 93 (2), 455-467.

Bochwic, T., 2000. Narodziny i DziałAlność Solidarności Oświaty i Wychowania 1980-1989. Tysol.

Bonsang, E., Adam, S., Perelman, S., 2012. Does retirement affect cognitive functioning? J. Health Econ. 31 (3), 490-501.

- Cameron, A.C., Gelbach, J.B., Miller, D.L., 2008. Bootstrap-Based Improvements for Inference with Clustered Errors. Rev. Econ. Stat. 90 (3), 414-427.
- Cantoni, D., Chen, Y., Yang, D.Y., Yuchtman, N., Zhang, Y.J., 2017. Curriculum and ideology. J. Polit. Econ. 125 (2), 338-392.

Cary, C.D., 1976. Patterns of emphasis upon Marxist-Leninist ideology: A computer content analysis of Soviet school history, geography, and social science textbooks. Comp. Educ. Rev. 20 (1), 11–29.

Costa-Font, J., Nicińska, A., 2023. Comrades in the family? soviet communism and demand for family insurance. Kyklos 76 (4), 526–612. Czapinski, J., 2011. Diagnoza Społeczna 2011 Warunki i Jakość Życia Polaków - cały raport (PL). Contemp. Econ. 5 (3). Czerwiec, E., 2009. Charakterystyka rent z tytułu niezdolności do pracy przyznanych w polsce w latach 2003–2006. Zesz. Naukowe Uniw. Ekon. Krakowie 808, 5–18.

Dahlum, S., Knutsen, C., 2017. Do democracies provide better education? Revisiting the democracy-human capital link. World Dev. 94.

- Davenport, C., 2007. State Repression and the Domestic Democratic Peace. Cambridge Studies in Comparative Politics.
- Davenport, C., Armstrong, D.A., 2004. Democracy and the violation of human rights: A statistical analysis from 1976 to 1996. Am. J. Polit. Sci. 48 (3), 538-554.

Diner, D., 2004. Remembrance and knowledge: Nationalism and stalinism in comparative discourse. In: The Lesser Evil. Routledge, pp. 93-105.

Diwan, I., Vartanova, I., 2020. Does education indoctrinate? Int. J. Educ. Dev. 78.

Dobosiewicz, S., 1971. Reforma Szkoł Y Podstawowej. Książka i Wiedza, Warsaw.

Eberstadt, N., 2003. Population aspects of communist countries. https://www.aei.org/articles/population-aspects-of-communist-countries/, Accessed: 2022-10-03. Fuchs-Schündeln, N., Masella, P., 2016. Long-lasting effects of socialist education. Rev. Econ. Stat. 98 (3), 428–441.

Gerovitch, S., 2007. "New soviet man" inside machine: Human engineering, spacecraft design, and the construction of communism. Osiris 22 (1), 135–157.

Goddeeris, I., 2008. Solidarnosc, the Western World, and the End of the Cold War. Eur. Rev. 16 (1), 55.

Góra, M., 1999. Ekonomiczne podstawy funkcjonowania nowego systemu emerytalnego w polsce. Gospod. Nar. (3), 9-24.

Góra, M., Krzak, M., Szymecka, A., 2008. Reforma Finansów Publicznych w Polsce. Center for Social and Economic Research (CASE), Warsaw.

Görlitz, K., Gravert, C., 2018. The effects of a high school curriculum reform on university enrollment and the choice of college major. Educ. Econ. 26 (3), 321-336.

Gorodnichenko, Y., Roland, G., 2011a. Individualism, innovation, and long-run growth. Proc. Natl. Acad. Sci. 108 (Supplement 4), 21316–21319.

Gorodnichenko, Y., Roland, G., 2011b. Which dimensions of culture matter for long-run growth? Am. Econ. Rev. 101 (3), 492-498.

Gorodnichenko, Y., Roland, G., 2012. Understanding the individualism-collectivism cleavage and its effects: Lessons from cultural psychology. In: Institutions and Comparative Economic Development. pp. 213–236.

Granato, J., Inglehart, R., Leblang, D., 1996. The effect of cultural values on economic development: Theory, hypotheses, and some empirical tests. Am. J. Polit. Sci. 40 (3), 607–631.

Hall, D.T., Yip, J., Doiron, K., 2018. Protean careers at work: Self-direction and values orientation in psychological success. Annu. Rev. Organ. Psychol. Organ. Behav. 5, 129–156.

Hansen, C.W., 2013. Economic growth and individualism: The role of informal institutions. Econom. Lett. 118 (2), 378-380.

Hoisington, T., Shabad, S., Siegelbaum, L.H., Sokolov, A., 2000. Stalinism as a Way of life: a Narrative in Documents. Yale University Press.

Huebener, M., Kuger, S., Marcus, J., 2017. Increased instruction hours and the widening gap in student performance. Labour Econ. 47 (C), 15-34.

Ignaczak, L., et al., 2019. Słowno-muzyczne widowisko składankowe jako palimpsest kulturowy-na przykładzie 'Niech no tylko zakwitną jabłonie' Agnieszki Osieckiej. Acta Univ. Lodz. Folia Litt. Pol. 52 (1), 425–445.

Jankowiak, S., 2001. Zaszczepić zasady socjalistycznej moralności. Biul. Inst. Pamięci Nar. 1 (10), 31-36.

Janowski, A., 1992. Polish education: Changes and prospects. Oxf. Stud. Comp. Educ. 2 (1), 41-55.

Janowski, A., 1999. The specific nature and objectives of civic education in Poland: Some reflections. In: Civic Education Across Countries: Twenty-Four National Case Studies from the IEA Civic Education Project. pp. 463–482.

Jarosz, E., 1998. Przemoc Wobec Dzieci: Reakcje śRodowisk Szkolnych. Wydawnictwo Uniwersytetu Śląskiego.

Jędrych, K., 2014. Lektury W Programach Dla SzkołY Podstawowej Z Lat 1949-1989. Wydawnictwo Uniwersytetu Śląskiego, Katowice.

Kikuchi, N., 2014. The effect of instructional time reduction on educational attainment: Evidence from the Japanese curriculum standards revision. J. Jpn. Int. Econ. 32, 17–41.

Krzywicki, A., 1955. Poststalinowski karnawał radości. V światowy Festiwal Młodzieży i Studentów o Pokój i Przyjaźń, Warszawa.

Kurian, G.T., 1988. World Education Encyclopedia. Vol. 3. Major Countries-Sudan-Zambia. Facts on File Publications, Middle Countries-Afghanistan-Zimbabwe. Minor Countries-Andorra-Yemen. Appendixes.

Lavy, V., 2010. Do differences in schools' instruction time explain international achievement gaps? Evidence from developed and developing countries. Econ. J. 125.

Lichter, A., Löffler, M., Siegloch, S., 2020. The long-term costs of government surveillance: Insights from Stasi spying in East Germany. J. Eur. Econom. Assoc., Lopez, C., Marlow-Ferguson, R., 2002. World Education Encyclopedia: A Survey of Educational Systems Worldwide. Gale Group.

Mader, W., 1988. Walka Solidarności O Społ Eczny Kształ Oświaty W Polsce. Polska Macierz Szkolna, London.

Mauersberg, S., 1996. Wykonywanie obowiązku szkolnego w niepodległej Polsce (1918-1939). Rozpr. Dziejów Oświaty 37, 155-176.

Mauersberg, S., Walczak, M., 2005. Szkolnictwo Polskie Po Drugiej Wojnie śWiatowej:(1944-1956). PTP ZG.

Myck, M., Nicinska, A., Morawski, L., 2009. Count Your Hours: Returns to Education in Poland. IZA Discussion Papers 4332, Institute of Labor Economics (IZA). Nikolova, M., Popova, O., Otrachshenko, V., 2019. Stalin and the Origins of Mistrust. Technical Report 344, Global Labor Organization (GLO).

Parinduri, R., 2014. Do children spend too much time in schools? Evidence from a longer school year in Indonesia. Econ. Educ. Rev. 41 (C), 89-104.

Pischke, J.-S., 2007. The impact of length of the school year on student performance and earnings: Evidence from the German short school year. Econ. J..

Rose, H., Betts, J.R., 2004. The effect of high school courses on earnings. Rev. Econ. Stat. 86 (2), 497-513.

Rzońca, A., Wojciechowski, W., 2008. Ile Kosztują nas Wcześniejsze Emerytury. Raport Forum Obywatelskiego Rozwoju, Warszawa, pp. 4-5.

Sieklucki, D., 2006. Partie Lewicy I Centrolewicy W Polskim Systemie Partyjnym: Aktywność SLD, PSL I UP Na Polskiej Scenie Politycznej. Wydawnictwo Uniwersytetu Jagiellońskiego, Kraków.

Slomczynski, K.M., Miller, J., Kohn, M.L., 1981. Stratification, work, and values: A Polish-United States comparison. Am. Sociol. Rev. 720-744.

Stańczyk, X., 2019. Stalinowska socjogeneza kultury młodzieżowej w PRL. Próba nowego ujęcia. Stud. Litt. Hist. (8).

Świda-Ziemba, H., 1998. My nowe życie tworzymy i nowy ład. In: Siciński, A. (Ed.), Do i od socjalizmu. PWN, Warsaw, pp. 91-109.

Szczepanek, M., 2018. Przemiany polskiego systemu edukacji podstawowej od lat 80. XX wieku. Eduk.-Tech.-Inform. 9 (4), 524–528.

Szebenyi, P., 1992. Change in the systems of public education in East Central Europe. Comp. Educ. 28 (1), 19-31.

Tomasik, W., 2016. inŻYnieria Dusz. Literatura Realizmu Socjalistycznego W Planie "Propagandy Monumentalnej". Wydawnictwo Naukowe Uniwersytetu Mikołaja Kopernika.

Torbert, W.R., 1978. Educating toward shared purpose, self-direction and quality work: The theory and practice of liberating structure. J. High. Educ. 49 (2), 109–135.

Viinikainen, J., Kokko, K., Pulkkinen, L., Pehkonen, J., 2010. Personality and labour market income: Evidence from longitudinal data. Labour 24 (2), 201–220. Wagner, B., 2018. Strategia Wychowawcza PRL. Prace Instytutu Historycznego UW.

Zysiak, A., 2016. Punkty Za Pochodzenie: Powojenna Modernizacja I Uniwersytet W Robotniczym MieśCie. Nomos.